Author's response to reviews

Title: Synchronously Diagnosed Eosinophilic Granuloma and Hodgkin Disease: a case report

Authors:

Soheila Sarmadi (ssarmadi@sina.tums.ac.ir)
Amir B Heidari (amirbehzad@yahoo.com)
Amir H Sina (amirhoseinsina@yahoo.com)
Mohamad A Ehsani (mohamadali.ehsani@yahoo.com)

Version: 2 Date: 15 April 2008

Author's response to reviews:

Dear sir

Referring to the letter 29 March 2008, I shall inform you that all revisions that were recommended in the letter were done by the authors including patients ethnicity and list of abbreviations. I would appreciate if you consider the manuscript for publishing.

Best Regards

Dr. Heidari